Author’s response to reviews

Title: SUrgical versus PERcutaneous Bypass: SUPERB-trial; Heparin-bonded endoluminal versus surgical femoro-popliteal bypass: study protocol for a randomized controlled trial

Authors:

Mare MA Lensvelt (m.m.a.lensvelt@gmail.com)
Suzanne Holewijn (SHolewijn@rijnstate.nl)
Wilbert M Fritschy (w.m.fritschy@isala.nl)
Otmar RM Wikkeling (m.wikkeling@nijsmellinghe.nl)
Laurens A van Walraven (l.vwalraven@antonius-sneek.nl)
Bas M Wallis de Vries (b.m.wallisdevries@chir.umcg.nl)
Clark J Zeebregts (czeebregts@hotmail.com)
Michel MPJ Reijnen (MReijnen@rijnstate.nl)

Version: 5 Date: 16 July 2011

Author’s response to reviews: see over
To
Doug Altman
Curt Furberg
Jeremy Grimshaw
Peter Rothwell
Editors-in-Chief
Trials

Date        July 16, 2011

Re.  'SUrgical versus PERcutaneous Bypass: SUPERB-trial; Heparin-bonded endoluminal versus surgical femoro-popliteal bypass; study protocol for a randomized controlled trial (MS 4436397405471142)'

Dear Editors,

Please find enclosed the revision of this manuscript, which incorporates the remaining changes that you advised.

Yours sincerely,

Clark J. Zeebregts, MD, PhD
Professor of Vascular Surgery
Department of Surgery, Division of Vascular Surgery
University Medical Center Groningen
P.O. Box 30001, 9700 RB Groningen
The Netherlands

Tel.: +31-50-3613382
Fax: +31-50-3611745
E-mail: czeebregts@hotmail.com